Author's response to reviews

Title: CD4 positive/CD8 negative/CD56 positive T cell Large Granular Lymphocyte Proliferations. Clonal Disorders of Uncertain Significance.

Authors:

Deepti Mutreja MD (deeptimutreja@gmail.com)
Rahul K Sharma MSc (rahulsharmabio@gmail.com)
Renu Saxena MD (renusaxena@outlook.com)

Version: 4 Date: 31 May 2014

Author's response to reviews: see over
**REVIEWER 1**

<table>
<thead>
<tr>
<th>REVIEWER’S COMMENTS</th>
<th>AUTHOR’S REPLY</th>
</tr>
</thead>
<tbody>
<tr>
<td>1. Authors must include in figures the results from TCR gamma rearrangement PCR. It is of interest to the reader to evaluate clonality.</td>
<td>Clonality has been ascertained on TCR gamma gene rearrangements PCR done from outside sources and not performed in our laboratory, hence the authors regret to inform that figures are not available.</td>
</tr>
<tr>
<td>2. The word &quot;commoner&quot; was not corrected as requested in the manuscript and was used twice in the resubmission.</td>
<td>“commoner” replaced by more common</td>
</tr>
</tbody>
</table>

**REVIEWER 2**

<table>
<thead>
<tr>
<th>REVIEWER’S COMMENTS</th>
<th>AUTHOR’S REPLY</th>
</tr>
</thead>
</table>
| 1. Major Compulsory Revisions  
Abstract: The information presented in the abstract is somewhat inconsistent and misleading, compared to the one listed in the Introduction, as it does not include the presence of a high anti-CMV IgG titre in both patients  
- Minor Essential Revisions  
1. Still requires some English language editing for uniformity throughout the manuscript. | Presence of a high anti-CMV IgG titre has been added to abstract |
| 2. Minor Essential Revisions  
1. Still requires some English language editing for uniformity throughout the manuscript | “commoner” replaced by more common  
“Fatigability” replaced by increased fatigue |